Author's response to reviews

Title: Adolescent pregnancies and girls' sexual and reproductive rights in the amazon basin of Ecuador: an analysis of providers’ and policy makers’ discourses

Authors:

Isabel Goicolea (isabel.goicolea@obgyn.umu.se)
Marianne Wulff (marianne.wulff@vll.se)
Miguel San Sebastian (miguel.sansebastian@epiph.umu.se)
Ann Öhman (ann.ohman@epiph.umu.se)

Version: 7 Date: 2 June 2010

Author's response to reviews:

Dear editor

Find attached the revised manuscript. We have revised the references as requested. Reference regarding sugar dadies (number 41 in the previous manuscript) has been removed, and references in the text follow consecutive numbers. Please let us know if there is any other change that we should made.

Kind regards,

Isabel Goicolea